Author's response to reviews

Title: PRIMARY MELANOMA OF THE PROSTATE

Authors:

Tosev MD Georgi (georgi.tosev@med.uni-heidelberg.com)
Kuru MD Timur (tkuru@ukaachen.de)
Johannes PD Huber (Johannes.Huber@uniklinikum-dresden.de)
Gerald MD Freier (gerald.freier@med.uni-heidelberg.de)
Frank MD Bergmann (frank.bergmann@med.uni-heidelberg.de)
Jessica MD Hassel (jessica.hassel@med.uni-heidelberg.de)
Sascha PD Pahernik (sascha.pahernik@uni-heidelberg.de)
Markus Prof. Hohenfellner (marjus.hohenfellner@med.uni-heidelberg.de)
Boris Prof. Hadaschik (boris.hadaschik@med.uni-heidelberg.de)

Version: 2
Date: 4 December 2014

Author's response to reviews: see over
30 November 2014

Dear editor,

Attached please find our manuscript “Primary melanoma of the prostate”.

In this case report we discuss treatment of a young patient with primary melanoma of the prostate. Radical prostatectomy was performed to prevent local complications such as lower urinary tract symptoms and hematuria. After systemic relapse immunooncologic treatment led to improved overall survival as compared to previous reports.

We believe that our case is of interest to the readers of Journal of BMC urology and look forward to your editorial comments.

Please do not hesitate to contact me in case of questions.

Sincerely,

Georgi Tosev